Author's response to reviews

Title: Shift Work to Balance Everyday Life? An interview study with a salutogenic perspective among registered and assistant nurses in home help service in Sweden

Authors:

Madelaine Agosti (madelaine.agosti@spray.se)
Ingemar Andersson (ingemar.andersson@hkr.se)
Göran Ejlertsson (work@ejlertsson.se)
Ann-Christin Janlöv (ann-christin.janlov@hkr.se)

Version: 2 Date: 15 July 2014

Author's response to reviews:

Dear Editor,

Thank you for your comments. We have made the changes that you requested.

- We have stated in our manuscript that this study was a part of a comprehensive PhD thesis work, ethically approved by the Local Ethical Review Board of Lund and funded by Kristianstad University.

- We have described; why 15 of 31 were contacted for study participation, why only one of 13 was an RN and why the pilot interview was included in the analysis.

Yours sincerely,

Madelaine Agosti